

# Title: Genetic map of Wilson disease in Spain - A great tool to improve diagnosis and screening

## Authors:

Pablo Alonso-Castellano, Zoe Mariño, Antonio Tugores, Antonio Olveira, Javier Ampuero , Marina Berenguer , José Ramón Fernández-Ramos, Jose María Moreno-Planas, María Lázaro-Ríos, Helena Masnou, María Luisa González-Diéguez, José María Pinazo-Bandera, Esther Molina-Pérez, Manuel Hernández-Guerra, Marta Romero-Gutiérrez, Paula Fernández-Álvarez, Carolina Muñoz , Sara Lorente, Alba Cachero, Manuel Delgado-Blanco, Víctor Vargas, Judith Gómez-Camarero, Francisca Cuenca, Luis Ibáñez-Samaniego, Miguel Fernández-Bermejo, Beatriz Álvarez-Suárez, Paula Iruzubieta, Ana Arencibia-Almeida, Luis García-Villarreal

DOI: 10.17235/reed.2025.11005/2024 Link: <u>PubMed (Epub ahead of print)</u>

Please cite this article as:

Alonso-Castellano Pablo , Mariño Zoe, Tugores Antonio, Olveira Antonio, Ampuero Javier, Berenguer Marina, Fernández-Ramos José Ramón, Moreno-Planas Jose María , Lázaro-Ríos María, Masnou Helena, González-Diéguez María Luisa, Pinazo-Bandera José María, Molina-Pérez Esther, Hernández-Guerra Manuel, Romero-Gutiérrez Marta, Fernández-Álvarez Paula, Muñoz Carolina, Lorente Sara, Cachero Alba, Delgado-Blanco Manuel, Vargas Víctor, Gómez-Camarero Judith, Cuenca Francisca, Ibáñez-Samaniego Luis, Fernández-Bermejo Miguel, Álvarez-Suárez Beatriz , Iruzubieta Paula , Arencibia-Almeida Ana, García-Villarreal Luis. Genetic map of Wilson disease in Spain -A great tool to improve diagnosis and screening. Rev Esp Enferm Dig 2025. doi: 10.17235/reed.2025.11005/2024.

This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

Revista Española de Enfermedades Digestivas The Spanish Journal of Gastroenterology

# Genetic map of Wilson Disease in Spain: a great tool to improve diagnosis and screening

- Wilson disease is a genetic disease and more than 1000 different mutations have been found in patients around the world
- We had no national map of the mutations affecting our patients and the current study presents the first genetic map of mutations for Wilson disease
- This is a very useful tool to improve early diagnosis of patients, what is crucial to offer them the best outcomes
- The widespread use of national maps will also increase our knowledge of the disease and allow the best design of multicenter trials in the future

Alonso-Castellano, et al.

Revista Española de Enfermedades Digestivas (REED) The Spanish Journal of Gastroenterology







Genetic map of Wilson disease in Spain - A great tool to improve diagnosis and screening

Authors:

1- **Pablo Alonso-Castellano\***. Universidad Las Palmas Gran Canaria. Servicio Digestivo, Complejo Hospitalario Universitario Insular Materno Infantil (CHUIMI), Las Palmas de Gran Canaria. pabloalonsoc1997@gmail.com

2- **Zoe Mariño\*.** Liver Unit, Hospital Clínic de Barcelona, IDIBAPS, CIBERehd, ERN-RARE Liver, Universitat de Barcelona, Barcelona. ZMARINO@clinic.cat

3- Antonio Tugores. Unidad de Apoyo Investigación. CHUIMI. Las Palmas Gran Canaria. uichuimi@gmail.com

4- Antonio Olveira. Digestive Diseases. Hospital Universitario La Paz, Madrid. aolveiram@gmail.com

5- Javier Ampuero. Digestive Diseases. Hospital Universitario Virgen del Rocío, Sevilla. jampuero-ibis@us.es

6- Marina Berenguer. Digestive Diseases. Hospital Universitari i Politècnic La Fe, València. marina.berenguer@uv.es

7- José Ramón Fernández-Ramos. Digestive Diseases. Hospital Universitario de Cruces, Baracaldo. joseramon.fernandezramos@osakidetza.eus

8- José María Moreno-Planas. Digestive Diseases, Complejo Hospitalario Universitario de Albacete, Facultad de Medicina Universidad de Castilla La Mancha. josemariamoren@yahoo.es

9- María Lázaro-Ríos. Hospital Universitario Miguel Servet, Zaragoza. marialazarorios@hotmail.com

10- Helena Masnou. Digestive Diseases. Hospital Universitario Germans Trias i Pujol, Badalona. hmasnou.germanstrias@gencat.cat

11- María Luisa González-Diéguez. Hospital Universitario Central de Asturias, Oviedo. luisagondi@hotmail.com

12- José María Pinazo-Bandera. Unidad de Hepatología, Unidad de Gestión Clínica de Aparato Digestivo, Hospital Universitario Virgen de la Victoria, Instituto de Investigación Biomédica de Málaga-Plataforma Bionand, Málaga.



Josepinazo@hotmail.es

13- Esther Molina-Pérez. Digestive Diseases. Hospital Clínico de Santiago, Santiago de Compostela. esther.molina.perez@sergas.es

14- Manuel Hernández-Guerra. Digestive Diseases. Hospital Universitario de Canarias, Santa Cruz Tenerife. mhernand@ull.edu.es

15- Marta Romero-Gutiérrez. Digestive Diseases. Hospital Universitario de Toledo, Toledo. m.romero.gutierrez@gmail.com

16- Paula Fernández-Álvarez. Gastroenterology and Digestive Endoscopy. Hospital Universitario Virgen Macarena, Sevilla. paulafer7@gmail.com

17- Carolina Muñoz. Hospital Universitario 12 de Octubre, Madrid. cmcodoceo@salud.madrid.org

18- Sara Lorente. Hepatology and Liver Transplant Unit, Hospital Clínico Lozano Blesa. Zaragoza. slorente@salud.aragon.es

19- Alba Cachero. Hospital Universitari de Bellvitge, Hospitalet de Llobregat. alba.cachero@bellvitgehospital.cat

20- Manuel Delgado-Blanco. Digestive Diseases. Hospital Universitario A Coruña, A Coruña. manuel.delgado.blanco@sergas.es

21- Víctor Vargas. Liver Unit. Hospital Vall d'Hebron, Universitat Autónoma Barcelona, CIBERehd, Barcelona. victor.vargas@uab.cat

22- Judith Gómez-Camarero. Digestive Diseases. Hospital Universitario de Burgos, Burgos. jgomcam@hotmail.com

23- Francisca Cuenca. Liver Unit, Digestive Diseases. Hospital Clínico San Carlos, Madrid. paquicuenca@yahoo.es

24- Luis Ibáñez-Samaniego. Digestive Diseases. Hospital General Universitario Gregorio Marañón, Madrid. luis.ibanez@salud.madrid.org

25- Miguel Fernández-Bermejo. Digestive Diseases. Hospital Universitario de Cáceres, Cáceres. mfbermejo@gmail.com

26- Beatriz Álvarez-Suárez. Digestive Diseases. Hospital Universitario Lucus Augusti (CHUL), Lugo. alvarezsuarezb@gmail.com

27- Paula Iruzubieta. Digestive Diseases. Hospital Universitario Marqués de Valdecilla, Santander. p.iruzubieta@gmail.com



28- Ana Arencibia-Almeida. Digestive Diseases. Hospital Universitario Nuestra Señora de La Candelaria, Santa Cruz de Tenerife. a.arencibiaa@yahoo.es
29- Luis García-Villarreal. Grupo de Investigación Patología Médica. Instituto de Investigaciones Biomédicas y Sanitarias. Universidad de Las Palmas de Gran Canaria. Digestive Diseases, Complejo Hospitalario Universitario Insular Materno Infantil (CHUIMI). Las Palmas de Gran Canaria. Igarciavillarreal@gmail.com

#### \*\* P A-C and ZM share first authorship

## Authors' contributions

LGV and AT were responsible for the study design. ZM, MB and P A-C, revised the manuscript for important intellectual content. All the authors were responsible for acquisition and curation of data and for revisión of the manuscript. All the authors have read and agreed to the published version of the manuscript.

## Corresponding author

Luis García-Villarreal. Grupo de Investigación Patología Médica. Instituto de Investigaciones Biomédicas y Sanitarias. Universidad de Las Palmas de Gran Canaria. Servicio Digestivo, Complejo Hospitalario Universitario Insular Materno Infantil (CHUIMI). Las Palmas de Gran Canaria. Tfno 928441618 Igarciavillarreal@gmail.com ORCID 0000-0002-6671-8061

#### List of abbreviations

WD: Wilson disease. AEEH: Spanish Association for Study of the Liver (Asociación Española para Estudio del Hígado. LOF: Loss of function. IQR: Interquartile range

## Ethics approval

The protocol of the study was reviewed and approved by all the Ethical Committees of the participating centers. All included subjects provided their informed consent. The research was carried out pursuant to the ethical principles established in the Declaration of Helsinki



#### ABSTRACT

**Background**: A National Registry for Wilson disease (WD) was recently started by the Spanish Association for Study of the Liver (AEEH). We evaluated the genetic data of the patients and the differences between regions and hospitals.

**Method**: Multicentric observational study from the WD Registry after the first year. **Results**: Patients from 30 hospitals, in 13/17 Spanish regions (covering 80% of population) were included. Genetic data were available for 260/320 patients. More than 130 mutations in the *ATP7B* gene were registered, the majority in less than 4 alleles, being most prevalent p.Met645Arg (20% alleles, mainland and the Canary Islands), p.Leu708Pro (16.5%, Canary Islands) and p.His1069Gln (8.3%, mainland). Only 15 mutations occurred in homozygosis, 3 in more than 5 patients: p.Leu708Pro (24 patients), p.Met645Arg and p.His1069Gln (6 patients each). Genetic data availability ranged from 0-100% among regions; similarly, the difference among hospitals within a region was larger than 50%. Without a genetic test up to 45% of patients would not have reached Leipzig score >3. In screening cases, genetics was used in 45/58 (78%); without genetics, two thirds of them would not have reached a Leipzig score >3 (33/49).

**Conclusion**: Here we present the first genetic map of WD in Spain. The use of genetic test was highly heterogeneous, being higher in screening cases. Although a large variability of mutations was found, regional characteristics indicated that screening for a limited number of exons (6, 8, 10, 14 or 17) would detect more than 50% of alleles in a given region, thus enabling the design of better diagnosis/screening tailored strategies.

Keywords: Genetic screening. Rare disease. ATP7B gene. Misdiagnosis.

## Lay Summary



Wilson disease (WD) is a rare genetic disease secondary to mutations in the *ATP7B* gene, encoding a protein dealing with copper excretion. There are more than 1000 different mutations among patients and diagnosis is sometimes difficult. Without treatment WD leads to death but there are therapies that are so effective to offer a normal life to patients, if started early. Every population shows different mutations, so a genetic map of every country is very important for patients and physicians, to reach early diagnosis and therapy with the best prognosis for patients.

#### **Conflict of interest**

VV: Speaker fees, Alexion Pharma, Consultant (advisory board), Ipsen

AM: speaker fees from Orphalan

ZM: speaker fees from Orphalan and Gilead; advisory fees from Orphalan, Alexion and Deep Genomics; grants from Gilead.

The others authors reported no competing interests

## INTRODUCTION

The Wilson disease (WD) is an autosomal recessive disorder of copper metabolism, caused by mutations in the *ATP7B* gene (1). Over 650 mutations are known to be pathogenic (2), according to the standards established by the American College of Medical Genetics and Genomics (ACMG) (3). While most reported mutations occur in single families, a few ones are typically frequent in certain populations.

The consequences of copper accumulation include the development of hepatic and/or neurological conditions of variable severity. However, consequences can be prevented by starting therapy at an early stage, even before the onset of symptoms. Therefore, knowing the geographical distribution of WD mutations is paramount to design appropriate screening strategies and to improve proper diagnosis in symptomatic cases. (4)

The prevalence of the disease is variable among regions. It has been estimated to be 1:29,000 to 1:40,000, although can be much higher in specific populations (5). Prevalence estimates based on the number of mutant allele carriers in the general



population predict much higher prevalence values, indicating that the disease might be underdiagnosed (6-12). Thus, genetic screening appears as a powerful tool when clinical hallmarks are not clearly defined for a given patient.

Genetic studies performed with central European patients have reported p.His1069Gln as the most prevalent mutation in WD patients, defining a general phenotype mostly based on this prevalent mutation. However, such studies did not include data from Spain (4), and preliminary data from earlier reports suggested that the prevalence in Spain could be different, since p.Met645Arg was found to be the most prevalent in WD patients in our country (13,14). The island of Gran Canaria is considered a WD hot spot with a strikingly high prevalence of the disease due to a specific mutation, p.Leu708Pro, associated with a founder effect (15).

The Spanish Association for the Study of the Liver (AEEH) recently established a registry of WD patients in our country, the AEEH WD Registry. This tool now offers the possibility to verify the true prevalences and establish a real map of genetic mutations in Spanish patients. On this background, the objectives of this study were:

- To describe the genetic characteristics of WD patients in Spain, showing most frequent mutations and their homozygote/heterozygote frequencies
- To evaluate the proportion and distribution of main mutation groups: Missense and Loss of Function.
- To evaluate regional differences in the occurrence of most prevalent mutations and in the use of genetic diagnosis.
- To identify exons carrying the highest proportion of mutations in patients from each region.
- To verify the feasibility and usefulness of genetic testing for diagnosis and screening of patients in our registry.

To the best of our knowledge, this is the first genetic analysis of ATP7B mutations covering almost all the regions of Spain. We postulate that our results could contribute to design the most suitable WD screening strategies for Spain, thus enhancing early diagnosis, which is critical in WD patients.



#### PATIENTS AND METHODS

#### Patients

This multicenter retrospective study was based on data of WD patients included in the AEEH-WD Registry between December 2021 (date of creation of the registry) and November 2022, attended at one of 28 public health care centers in Spain (covering approximately 80 % of the population). The protocol of the study was reviewed and approved by all the Ethical Committees of the participating centers. All patients included in the Registry had provided written informed consent to the use of their clinical data for research purposes.

## **Clinical and biochemical parameters**

Patient data were collected from Hospitals' registries on a retrospective basis. Patients were scored following the Leipzig criteria (16). We collected their clinical, demographic, analytical and genetic information.

#### **Genetic analysis**

Genetic analysis to search for *ATP7B* gene variants included double pass heterozygous Sanger sequencing of *ATP7B* coding and flanking sequences, and MLPA analyses when appropriate and available. Next-generation-sequencing was not used. The Ensembl database was used for identification of already known mutations. Missense variants were defined as a substitution in a coding base pair predicted to result in an amino acid change. Loss-of-function (LOF) variants were defined as nonsense (substitution in a coding base pair predicted to result in a premature stop codon), frameshift (insertion or deletion predicted to result in a shift in the reading frame) and splice site (substitutions predicted to involve a splice site resulting in splicing abnormalities).

## RESULTS

Baseline patients' characteristics



One year after its creation, the registry included a total of 320 WD patients (with Leipzig score >3), from 14 different Spanish regions, whose health services covered approximately 40 million people; 167 were men (52.2%) and the median age at diagnosis was 16 years (IQR<sub>25-75</sub> 10-27.8). One third of patients (32.2%) reported a positive family history. The predominant phenotype at diagnosis was chronic liver disease (n=158, 49%), followed by pre-symptomatic cases (n=66, 20.6%), neurological presentations (n=32, 10.1%), mixed phenotypes (n=29, 9.1%) and acute liver disease (n=29, 9.1%).

Genetic data were available for 260 out of 320 patients (81%), including 66 homozygotes, 154 compound heterozygotes and 40 patients with only one mutation. The number of patients from different regions was quite heterogeneous, being higher in Catalonia (80 patients) and the Canary Islands (79 patients), and very low in Extremadura and Cantabria (1 patient each). Regions with more than one participating hospital also showed high internal variability: Catalonia ranged 4-63 patients per center and the Canary Islands, 1-71 per center (Table 1).

The rate of utilization of genetic testing was very variable among the different regions of Spain; it was more used in the Valencia Community and Navarra (100% both) and poorly used in Cantabria and Extremadura. Within regions with more than one participating hospital the use was also highly variable, with differences among centers in more than 80% of cases in Madrid or Canary Islands, for instance (Table 1).

#### Observed mutations, frequencies and homozygosity

After examining 448 alleles, 130 different mutations were found, most of them (91%) affecting less than 4 alleles. The most frequent mutations were: p.Met645Arg (92 alleles i.e. 20% alleles, found both in mainland Spain and the Canary Islands), p.Leu708Pro (75 alleles, 16.5%, only in the Canary Islands), p.His1069Gln (38 alleles, 8.3%,only in mainland Spain), promoter region (14 alleles, 3.3%, both in mainland Spain and the Canary Islands) and c.1708-1G>A (12 alleles, 2.6%, only in mainland Spain). Fifteen mutations appeared in homozygosis, but only four of them occurred in more than 3 patients: p.Leu708Pro (n=24), p.Met645Arg (n=6), p.His1069Gln (n=6) and c.1708-1G>A (n=5), the latter one restricted to patients belonging to the gipsy ethnicity



from mainland Spain.

## Type of mutations

In terms of the type and localization of genetic alterations, most of them (74%) corresponded to missense mutations (332/448 alleles), while nonsense, deletion, insertion, splice sites and frameshift mutations affected 14% of alleles (61/448). Outside the coding regions 14 mutated alleles were found at the promoter region and 41 at introns: intron 1: n=8; intron 3: n=3; intron 4: n=15; intron 9: n=1; intron 13: n=11; intron 14: n=1; intron 17: n=2).

## Localization along exons

The distribution of mutations along the DNA sequence was also very heterogeneous, with only 6 out of the 21 exons presenting more than 10 different mutations: exon 8 (109/448: 24%, mainly p.Leu708Pro occurring in 80 alleles), exon 6 (93/448: 20% with 92 alleles presenting p.Met645Arg), exon 14 (54/448: 12%, 38 alleles with p.His1069Gln), exon 15 (31/448: 7%), exon 2 (24/448:5%) and exon 19 (12/448: 3%). No mutations were found at exons 3 and 21.

## Geographical distribution of the most prevalent mutations and affected exons

The distribution of the most frequent mutations was not homogeneous throughout the country, as shown along with the number of cases detected (Table 2). The most prevalent mutation in Spain, p.Met645Arg, was found in all regions, though with different frequencies, more abundant in regions away from the seashore (Figure 1A), while p.Leu708Pro was exclusively found in the Canary Islands, mainly in Gran Canaria, and p.His1069Gln was mainly found in northern regions (31/38 alleles, 82%) (Figure 1B). The change affecting the promoter region (c.-436-422del15) was restricted to the regions along Mediterranean coast and the Canary Islands. Mutation c.1708-1G>A was only found in the regions along Mediterranean coast and the Saturd was restricted to people of gypsy ethnicity. Also shown is the geographic distribution of most affected exons. As it can be observed, in many regions, 2 or 3 exons include more than 50% of mutations in WD patients (Table 2).



#### Value of genetic data for diagnosis

A review of data from registered patients with available genetic information (260 of 320 patients) showed that, if the score points obtained from genetic information were eliminated from the diagnostic procedure, almost half of them (126/260) would not have reached a Leipzig score higher than 3.

In the sub-group of patients who had been diagnosed after family screening, 78% of them had undergone a genetic test. If such information were removed, 60% would have showed a Leipzig score lower than 4. Conversely, only 130 (50%) out of the 260 patients with genetic information available, had undergone histologic evaluation. If biopsy information was removed and genetic testing kept, most of them (118/130: 90%) still had a Leipzig score higher than 3.

#### DISCUSSION

Several studies on WD in Spain including genetic data been published (13-15, 17) but information covering almost the entire country was not available until now, and allowed us to produce the first WD genetic map of Spain.

Firstly, the study shows that the use of genetic testing is not homogeneous among the regions that participated in the study. The reasons behind such variability are not clear. The tradition of the management of WD may be an important one. The size of the city may be a further factor influencing such heterogeneity, since hospitals in smaller cities often need to send their samples to other hospitals in larger cities. Many cases were diagnosed by clinical criteria only, so genetic data were not necessary. However, there was also a great variability between centers within the same region, suggesting that genetic testing accessibility did not seem to depend on autonomic region's diagnostic strategies, but rather linked to specific procedures adopted at each center, probably based on local expertise stressing the value of genetic testing. As it becomes cheaper, it seems that test price might not be as important as stressing its value. Indeed, we show that genetic testing was of great value during diagnosis, even more than intensively invasive procedures. We consider that the work presented will induce the



universal use of genetic testing for WD patients as a powerful tool to increase our knowledge and contribute to improve treatment and prognosis of patients.

The increasing use of total genome studies might reveal individuals with mutations in both alleles, who however remain asymptomatic or show atypical manifestations (21-23). The sooner we gain information on all possible mutations, the sooner we will be able to provide all patients with suitable diagnosis and treatment.

A strategy that may even make genetic testing more accessible is the customization to the screening process according to the territory being under scrutiny. Certain genetic variants are linked to specific populations, and that is also the case for disease causing mutations, sometimes simplifying genetic screening (4, 24). This first Spanish countrywide WD genetic evaluation revealed strikingly a high variability of disease causing mutations: Out of 448 alleles tested, up to 130 different mutations were found. On the other hand, we found remarkable homogeneity in certain subpopulations, which may evidence a founder effect. This is the case of Gran Canaria, where the mutation p.Leu708Pro is highly prevalent with a well-known founder effect dating several centuries back (15), or the intron 4 mutation (c.1708-1G>A), restricted to gypsy families along the Mediterranean coastline. Finally, mutation c.-436-422del15 affecting the promoter sequence, originally from Sardinia (24), was only found in the regions along the Mediterranean coast and in Gran Canaria, especially in families with ancestry on the most eastern islands Fuerteventura and Lanzarote (12) revealing specific population movements, since Sardinia was a part of the Spanish Kingdom for several decades and the commercial maritime exchange with the Spanish Mediterranean coast was intense. It should be noticed that this mutation has not been included in routine genetic testing for some years, so this might explain cases with only one or no mutant alleles when mutations were only looked at within exons and immediate flanking regions. Therefore, the extent of the Sardinian promoter mutation might may be redefined in the future as more data are gathered.

Our results confirmed that mutation p.Met645Arg, found both in mainland Spain and the Canary Islands, is the most prevalent one in our country (13, 18), with a great number of carriers in the population (12). According to global databases, this variant is poorly represented in other European populations but its frequency is remarkably



higher in Ashkenazis (20). The later finding might be related to the long-lasting presence several centuries ago of an important Jewish population in Spain, mainly Sephardic, but probably sharing common ancestors with Ashkenazi.

The pathogenicity of the p.Met645Arg mutation has been evaluated in several articles. In the first one (13), the authors suggested low pathogenicity, since they could not find homozygous patients. However, the registry included 6 homozygotes, with clinical and analytical confirmation. In a recent study, a second pathogenic mechanism, involving the splicing of the *ATP7B* transcript, has been proposed for this mutation (19). Its high allele frequency, both in the general Spanish population (12) and in the WD patient registry, should lead clinicians to consider it in WD diagnosis or in any patient presenting with liver disease of unknown etiology.

Our data confirmed the occurrence in our country of the most prevalent mutation in Eastern Europe, p.His1069Gln (4), which might be due to immigration from those countries, something reinforced by the fact that it was mainly concentrated in Northern regions and was restricted to Spain mainland territory.

In summary, genetic testing in Spain might be efficiently reduced to a few most prevalent variants, detecting most of the cases. These variants are found in a limited number of exons, making a first pass screening approach sensibly cheaper, considering that the most accepted screening technique is Sanger sequencing of individual exons. This is especially important as access to genetic testing, with results reaching almost the 100% of alleles in some centers, may change the WD diagnosis process. Performing a liver biopsy to measure hepatic copper content is an invasive and expensive procedure, often with inconclusive results due to the heterogeneous copper tissue distribution (25). Evaluating fibrosis with not invasive tools like elastography (26), could be a better follow-up method when properly evaluated. As expected, the most widespread use of genetic testing concerns family screenings where only the mutations of the index case are looked for. However, this can only be done if genetic tests have been previously carried out inpatients, which is a further reason to recommend a wider adoption of genetic studies.

The main limitations of the present work were, on the one hand, reduced recruitment of both patients and hospitals and, on the other hand, the lack of genetic information



from approximately 25% of the patients included. However, it should be noticed that the largest earlier WD national study (27) included less patients and did not address genetic information. Another issue to consider when evaluating the number of cases diagnosed per region is the fact that communities such as Catalonia, Galicia or the Valencia Community receive samples to genotype from other regions lacking resources, a fact that is known but not finally registered in the database.

In conclusion, this study offers the first nationwide genetic map of Wilson disease in Spain, which may contribute to improve the diagnosis of future cases. The different mutations are depicted with their particularities throughout regions, which could help reduce the costs of genetic testing; namely, although *ATP7B* is a large gene with 21 exons, we showed that over 50% of mutations can be detected by analyzing 2 to 3 exons, which reduces costs and increases feasibility (6). Our results showed that genetic diagnosis can be better tailored to particular target populations, something considered of paramount importance in genetic screening, especially in cases like WD, involving almost 1000 possible mutations.

WD is a well known severe disease, for which an effective therapy is available, that has serious consequences, leading to irreversible damage, preventable through early diagnosis. The two main factors influencing the cost-effectiveness of a screening protocol are the cost of the diagnostic test and the prevalence of the disease: first, genetic test procedures have progressed remarkably, with the consequent reduction in prices, which could be even lower in the future; second, WD prevalence has been largely debated in recent years, with conflicting results from clinical versus genetic studies, the former yielding lower prevalences than the latter (6-12). One reason might be the low penetrance of certain frequent mutations, like p.Met645Arg in Spain (28), a finding that emphasizes the importance of large registries, like the AEEH WD Registry, for more accurate prevalence estimations, associated to the publication of the corresponding genetic map in every country to facilitate genetic testing in every suspected case.

#### REFERENCES



1. Członkowska A, Litwin T, Dusek P, et al. Wilson disease. Nat Rev Dis Primers. 2018 Sep 6;4(1):21

2. Kumar M, Gaharwar U, Paul S, et al. WilsonGen a comprehensive clinically annotated genomic variant resource for Wilson's Disease. Sci Rep. 2020 Jun 3;10(1):9037.

3. Richards S, Aziz N, Bale S, et al. Standards and guidelines for the interpretation of sequence variants: a joint consensus recommendation of the American College of Medical Genetics and Genomics and the Association for Molecular Pathology. Genet Med. 2015 May;17(5):405-24.

4. Ferenci P. Regional distribution of mutations of the ATP7B gene in patients with Wilson disease: impact on genetic testing. Hum Genet. 2006 Sep;120(2):151-9.

5. Sandahl TD, Laursen TL, Munk DE, Vilstrup H, Weiss KH, Ott P. The Prevalence of Wilson's Disease: An Update. Hepatology. 2020 Feb;71(2):722-732.

6. Coffey AJ, Durkie M, Hague S, et al. A genetic study of Wilson's disease in the United Kingdom. Brain 2013; 136:1476-87

7. Jang JH, Lee T, Bang S, et al. Carrier frequency of Wilson's disease in the Korean population: a DNA-based approach. J Hum Genet. 2017; 62:815-818.

8. Collet C, Laplanche JL, Page J, et al. High genetic carrier frequency of Wilson's disease in France: discrepancies with clinical prevalence». BMC Medical Genetics 2018; 19:143.

9. Gao J, Brackley S, Mann JP. The global prevalence of Wilson disease from next-generation sequencing data. Genet Med. 2019; 21:1155-1163.

10. Wallace DF, Dooley JS. ATP7B variant penetrance explains differences between genetic and clinical prevalence estimates for Wilson disease. Hum Genet. 2020; 139:1065-1075

11. Yamaguchi H, Nagase H, Tokumoto S, et al. Prevalence of Wilson disease based on genome databases in Japan. Pediatr Int. 2021; 63:918-922.

12. Lorente-Arencibia P, García-Villarreal L, González-Montelongo R, et al. Wilson Disease Prevalence: Discrepancy Between Clinical Records, Registries and Mutation Carrier Frequency. J Pediatr Gastroenterol Nutr. 2022; 74:192-199.



13. Margarit E, Bach V, Gómez D, et al. Mutation analysis of Wilson disease in the Spanish population -- identification of a prevalent substitution and eight novel mutations in the ATP7B gene. Clin Genet. 2005 Jul;68(1):61-8.

14. Brage A, Tomé S, García A, et al. Clinical and molecular characterization of Wilson disease in Spanish patients. Hepatol Res. 2007 Jan;37(1):18-26.

15. García-Villarreal L, Daniels S, Shaw SH, et al. High prevalence of the very rare Wilson disease gene mutation Leu708Pro in the Island of Gran Canaria (Canary Islands, Spain): a genetic and clinical study. Hepatology. 2000 Dec;32(6):1329-36.

16. Ferenci P, Caca K, Loudianos G, et al. Diagnosis and phenotypic classification of Wilson disease. Liver Int. 2003 Jun;23(3):139-42.

17. Huarte-Muniesa MP, Lacalle-Fabo E, Uriz-Otano J, et al. Dificultades en el diagnóstico de los pacientes con enfermedad de Wilson en la práctica clínica: experiencia de 15 casos [Complexity of the diagnosis of Wilson disease in clinical practice: our experience in 15 patients]. Gastroenterol Hepatol. 2014 Aug-Sep;37(7):389-96.

18. García-Villarreal L, Hernández-Ortega A, Sánchez-Monteagudo A, et al. Wilson disease: revision of diagnostic criteria in a clinical series with great genetic homogeneity. J Gastroenterol. 2021 Jan;56(1):78-89.

19. Merico D, Spickett C, O'Hara M, et al. ATP7B variant c.1934T > G p. Met645Arg causes Wilson disease by promoting exon 6 skipping. NPJ Genom Med. 2020 Apr 8; 5:16.

20. Gudmundsson S, Singer-Berk M, Watts NA, et al. Variant interpretation using population databases: Lessons from gnomAD. Hum Mutat. 2022 Aug;43(8):1012-1030.

21. Arruda WO, Munhoz RP, de Bem RS, et al. Pathogenic compound heterozygous ATP7B mutations with hypoceruloplasminemia without clinical features of Wilson's disease. J Clin Neurosci. 2014; 21:335-336.

22. Tomiyasu K, Oshima T, Yoshii M, et al. [Wilson's disease presenting as Asperger syndrome]. Rinsho Shinkeigaku. 2019 Sep 25;59(9):589-591.

23. Wattacheril J, Shea PR, Mohammad S, et al. Exome sequencing of an adolescent with nonalcoholic fatty liver disease identifies a clinically actionable case of Wilson disease. Cold Spring Harb Mol Case Stud. 2018 Oct 1;4(5): a003087.



24. Zappu A, Magli O, Lepori MB, et al. High incidence and allelic homogeneity of Wilson disease in 2 isolated populations: a prerequisite for efficient disease prevention programs. J Pediatr Gastroenterol Nutr. 2008 Sep;47(3):334-8.

25. Ferenci P, Steindl-Munda P, Vogel W, et al. Diagnostic value of quantitative hepatic copper determination in patients with Wilson's Disease. Clin Gastroenterol Hepatol. 2005 Aug;3(8):811-8.

26. Loomba R, Adams LA. Advances in non-invasive assessment of hepatic fibrosis. Gut. 2020 Jul;69(7):1343-1352.

27. Berenguer M, Vergara M, Almohalla C, et al. Significant heterogeneity in the diagnosis and long-term management of Wilson disease: Results from a large multicenter Spanish study. Gastroenterol Hepatol. 2023 Oct;46(8):577-584.

28. Alonso-Castellano P, Tugores A, Mariño Z, et al (Spanish Wilson Registry AEEH). Low penetrance of frequent ATP7B mutations explains the low prevalence of Wilson disease. Lessons from real-life registries. Dig Liver Dis. 2024 Sep 24:S1590-8658(24)01000-4.



## Figure legend



Figure 1. Geographical distribution of the most prevalent ATP7B mutations in Spanish patients. Shown are the percentages of alleles with the p. Met645Arg (A) and p.His1069Gln (B) mutations in the Spanish regions. Some have been united due to the low number of patients. Ast-Cant: Asturias-Cantabria; BC: Basque Country; Val Com: Valencian Community. Basic map by Daniel Dalet provided by https://d-maps.com/.